Author's response to reviews

Title: Correlation between delayed-enhancement magnetic resonance and nitrate myocardial Tc-99m tetrofosmin scintigraphy in myocardial infarction: a case report

Authors:

Mauro Feola (m_feola@virgilio.it)
Gian Luca Rosso (rossogl@tiscali.it)
Alberto Biggi (biggi.a@ospedale.cuneo.it)
Stephane Chauvie (chauvie.s@ospedale.cuneo.it)
Giovanni Leonardi (leonardi.g@ospedale.cuneo.it)
Franca Margaria (margaria.f@ospedale.cuneo.it)
Maurizio Grosso (grosso.m@ospedale.cuneo.it)
Valeria Ferrero (valeria.ferrero@hotmail.com)

Version: 5 Date: 24 October 2007

Author's response to reviews: see over
Dear Editor,

please receive the final version of our manuscript entitled "Correlation between delayed-enhancement magnetic resonance and nitrate myocardial Tc-99m tetrofosmin scintigraphy in myocardial infarction: a case report." to be considered for publication as Case Report. I précised the complete affiliation of all authors; the emails of all the authors has been written on the first page. The references contained all the authors' name and not only the first six following by et al. In the paragraph ‘Authors’ contributions' was indicated that all authors read and approved the final version of the manuscript and finally images were cropped.

Please do not hesitate to contact me should you require any further information.

Yours sincerely,

Mauro Feola, MD.

Telephone number: +39-0171441646.
Fax number: +39-0171699545.
e-mail: m_feola@virgilio.it